 Vc/17-May-05
Week 17  starting 25 April 05
SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday  18 April 05 @ 4/25/2005 2:21:38 PM   
	4
	A1
	MBBR3331
	PT 12 Warm Up
	235.47

	7
	A2
	MBBR3293
	PT 13.2 Resist. Meas. 
	297.59

	8
	B1
	[image: image1.png]



	[image: image2.png]



	294.73

	5
	B2
	SSS079
	Prep 5 Cool Down
	4.47 HV

	1
	C1
	MBBL1127
	PT 9 CDAP HV
	1.90

	11
	C2
	MBBL1105
	PT 13.1 IAP @ Warm
	299.65

	9
	D1
	SSS175
	Prep 0 Scan Bench
	294.55

	10
	D2
	[image: image3.png]



	[image: image4.png]



	294.55

	6
	E1
	MBBR3317
	ICS 4 Final connection
	295.18

	12
	E2
	[image: image5.png]



	[image: image6.png]



	296.97

	2
	F1
	MBBR3310
	PT 12 Warm Up
	11.53

	3
	F2
	MBAL2171
	PT 3 QH Measure
	4.66 HV


Temperature data are taken from TBSM18 site 
Coordinator this week: N Smirnov; MTM support on call :   16-3602 
16  complete Magnet Tests cycles   ()
11  dipoles( 1 with MM)+ 5  SSS( 1  with MM) +  0  SSSS (with MM)
   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· Situation on Repair :   J Halik
· General Points 
· AOB

	Magnets under Test from 17-04-2005 to 23-04-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	#
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	

	1
	TBA1
	MBBR3316 
	[image: image7.png]



	


	13-04-2005 
	1 
	24 h.
	27 h.
	37 h.
	12 h.
	22 h.
	18-04-2005 
	122 h.
	Stripping
	

	2
	TBA1
	MBBR3331 
	[image: image9.png]



	


	18-04-2005 
	2 
	30 h.
	22 h.
	37 h.
	15 h.
	0 h.
	23-04-2005 
	104 h.
	Thermal cycle
	

	3
	TBA2
	MBAL2169 
	[image: image11.png]



	


	15-04-2005 
	1 
	50 h.
	23 h.
	23 h.
	16 h.
	10 h.
	20-04-2005 
	122 h.
	Stripping
	

	4
	TBB1
	SSS170 
	[image: image13.png]



	


	15-04-2005 
	1 
	27 h.
	15 h.
	45 h.
	8 h.
	21 h.
	20-04-2005 
	116 h.
	Stripping
	

	5
	TBB2
	SSS057 
	[image: image15.png]



	


	20-04-2005 
	1 
	29 h.
	11 h.
	20 h.
	9 h.
	14 h.
	23-04-2005 
	83 h.
	Stand By
	

	6
	TBB2
	SSS130 
	[image: image17.png]



	


	15-04-2005 
	1 
	22 h.
	18 h.
	38 h.
	15 h.
	18 h.
	20-04-2005 
	111 h.
	Stand By
	

	7
	TBC1
	MBBL1111 
	[image: image19.png]



	


	17-04-2005 
	1 
	32 h.
	35 h.
	25 h.
	14 h.
	11 h.
	21-04-2005 
	117 h.
	Stripping
	

	8
	TBC2
	MBBL1218 
	[image: image21.png]



	


	16-04-2005 
	1 
	25 h.
	48 h.
	17 h.
	17 h.
	15 h.
	21-04-2005 
	122 h.
	Stripping
	

	9
	TBD1
	SSS151 
	[image: image23.png]



	


	11-04-2005 
	1 
	29 h.
	26 h.
	103 h.
	9 h.
	77 h.
	21-04-2005 
	244 h.
	Stripping
	

	10
	TBD2
	SSS095 
	[image: image25.png]



	


	14-04-2005 
	2 
	0 h.
	97 h.
	9 h.
	16 h.
	30 h.
	20-04-2005 
	152 h.
	Stripping
	

	11
	TBE1
	MBAL1177 
	[image: image27.png]



	


	12-04-2005 
	1 
	25 h.
	33 h.
	32 h.
	13 h.
	10 h.
	17-04-2005 
	113 h.
	Stripping
	

	12
	TBE1
	MBBL2164 
	[image: image29.png]



	


	17-04-2005 
	2 
	28 h.
	27 h.
	37 h.
	13 h.
	0 h.
	22-04-2005 
	105 h.
	Thermal cycle
	

	13
	TBE2
	MBBL1216 
	[image: image31.png]



	


	15-04-2005 
	1 
	40 h.
	23 h.
	51 h.
	15 h.
	9 h.
	20-04-2005 
	138 h.
	Stripping
	

	14
	TBF1
	MBAR3329 
	[image: image33.png]



	


	17-04-2005 
	1 
	32 h.
	28 h.
	33 h.
	20 h.
	20 h.
	22-04-2005 
	133 h.
	Stripping
	

	15
	TBF2
	MBAL3162 
	[image: image35.png]



	


	13-04-2005 
	1 
	25 h.
	26 h.
	26 h.
	13 h.
	10 h.
	18-04-2005 
	100 h.
	Stripping
	

	16
	TBF2
	MBBR3242 
	[image: image37.png]



	


	18-04-2005 
	1 
	29 h.
	26 h.
	22 h.
	17 h.
	7 h.
	22-04-2005 
	101 h.
	Stripping
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR3316
	11608 A.
(D1-U)
	11727 A.
(D2-L)
	11889 A.
(D2_L)
	12240 A.
(D2 - L)
	12463 A.
(D2_L)
	12516 A.
(D2-L)
	12669 A.
(D2-L)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-

	2
	MBBR3331
	9139 A.
(D1_U)
	10656 A.
(D1-L)
	11252 A.
(D1-U)
	11536.9 A.
(D2-U)
	11677 A.
(D1_L)
	12094 A.
(D1-L)
	12352 A.
(D1-L)
	12491 A.
(D1_L)
	40 A.
(No Data)
	12761 A.
(D2_U)
	[image: image40.png]



	-
	-

	3
	MBAL2169
	10820 A.
(D2-U)
	12137 A.
(D2 - U)
	12507 A.
(D1_L)
	[image: image41.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS170
	12008 A.
(QI_34)
	12266 A.
(QE-34)
	12165 A.
(QE_34)
	[image: image42.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS057
	12553.7 A.
(QI_34)
	12837.6 A.
(QE-34)
	12697 A.
(QI-34)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS130
	10210 A.
(QE_12)
	11361 A.
(QE_12)
	12223 A.
(QE-12)
	12497 A.
(QI-12)
	12780 A.
(QI_34)
	12845 A.
(QI_12)
	12850 A.
[image: image44.jpg]



	12850 A.
[image: image45.jpg]



	-
	-
	-
	-
	-
	-

	7
	MBBL1111
	11781 A.
(D2 - L)
	12137 A.
(D2 - U)
	12261 A.
(D1_L)
	[image: image46.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	8
	MBBL1218
	12411 A.
(D1-U)
	12840 A.
(D1_L)
	12850 A.
[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	9
	SSS151
	12389 A.
(QI - 12)
	12436 A.
(QI-12)
	12832.3 A.
(QI-34)
	12850 A.
[image: image48.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	10
	SSS095
	12630.2 A.
(QI-34)
	12850 A.
[image: image49.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	11
	MBAL1177
	11798 A.
(D2 - U)
	12313 A.
(D1 - L)
	12856 A.
[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	12
	MBBL2164
	10899 A.
(D1-U)
	11473 A.
(D1-L)
	12179 A.
(D1 - U)
	12747 A.
(D1 - L)
	12848 A.
(D1 - L)
	12443 A.
(D1-L)
	12263 A.
(D1 - L)
	12240 A.
(D1_L)
	11947.7 A.
(D1-L)
	11712 A.
(D1-L)
	[image: image51.png]



	-
	-
	

	13
	MBBL1216
	11359 A.
(D2-U)
	11838 A.
(D2-U)
	12114 A.
(D1 - U)
	12319 A.
(D2 - L)
	12400 A.
(D2_L)
	12546 A.
(D1-U)
	12737 A.
(D2-U)
	12841 A.
(D2 - L)
	12850 A.
[image: image52.jpg]



	-
	-
	-
	-
	

	14
	MBAR3329
	10744 A.
(D1-L-S01)
	12313 A.
(D1-L-S01)
	12617 A.
(D1-L-S09)
	[image: image53.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	15
	MBAL3162
	9081 A.
(D2_L)
	11344 A.
(D1-U)
	12583 A.
(D1-U)
	12850 A.
[image: image54.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	16
	MBBR3242
	11966 A.
(D1-U)
	12013.6 A.
(D2-U)
	12133.8 A.
(D2-L)
	[image: image55.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	


Week 17  :  17 April to 23 April 05  (19 Magnet Tests)
[   3636 On bench rapid Thermal cycle   ]
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _19_M


	TBA1


	Magnet MBBR3331, a NC should be open because bar code on the magnet is written MBAR3331.
	
	
	

	Tuesday _19_M


	TBC2
	MB1216:  HV failed between Dipole-GND even with eth IFS box changed.
	MTM
	
	Federico called. The 1.5 kA quenches and the 3 kA quench were done.

Test repeated then OK.

	Tuesday _19_M


	TBA1
	Some flat signals during IAP @ warm
	MTM
	
	Federico changed a faulty potaim-card.

	Wednesday_20_M


	TBF1
	MM 1 on MB3229.Check by Stephane: polarity error message on one head of aperture 2.
	MTM
	
	Stephane found that the switch in the amplifier of the coil 0001 was in e position and the setting in a position. MM1 repeated.

	Wednesday_20_N


	TBD2
	IAP@warm. Trigger present on VClampE.
	ICS?
	
	Federico asked ICS to check the  SSS connection. OK after the reconnection of the magnet.

	Wednesday_20_N


	TBD2
	Upgrade of the MTF during the night. Trouble in the HV recognition system

Problem remains in the Thursday morning.
	TS/CE
	MTF
	HWREC switched off. Magnet mane entered manually.

Giorgio switched off HWREC of TBE and TBB, copied the config files in order to allow OP team to go on with the tests.

	Wednesday_20_N


	TBA1
	Ext2 interlock on RMA. It could not be reset.
	AB/PO
	
	Federico found a fault inside the discharge switch and has been reseted.

A reset of the communication card of RMA was also performed.

	Thursday_21_N

Friday_22_M

Sunday _25_M
	TBA1
	MBB3331:Faulty quench twice at very low current (40 A approx.). Trigger in Vbus-Ext, Vbus-Int.

Idem

Idem
	AB/PO
	
	Federico solved the problem by resetting the communication card on RMA

	Friday _22_A

	TBA1
	WP07 on MB not found
	ICS
	
	ICS team entered the file result wit the wrong magnet name..

	Friday _22_A

	all
	Server problem : MTF is not working
	IT
	MTF
	Tests performed with HWREC off. 

	Friday _22_A

	TBB2
	Warning for temperature sensor discrepancy found during the second training quench of SSS057.
	MTM
	
	Federico will check the coefficients of the temperature sensor.

	Friday _22_N

	TBB1
	Magnet name error in E-traveller done by ICS

SSS96 instead SSS096
	ICS
	
	

	Friday _22_N

	TBB1
	In PT5.2 provoked quench at 1.5 kA, symmetrical quench is detected, but voltage tap signals not reaching 200mV.
	MTM
	hf
	DAQ triggered on a spike.

	Friday _22_N

	TBA2
	During IAP @ cold, not possible to arm HF DAQ. 
	AB/CO
	hf
	

	Saturday_23_M

	TBA2
	During IAP @ cold on MB3293, few offset of QLoc cards were not very stable, a warning appeared

AC compensation signals not found
	MTM
	IAP
	Federico checked and  advised to proceed with the tests.

P20 connector not connected to the IFS box 

	Saturday_23_A

	TBC2
	Cryo problem on MBBL1105? 
	ACR
	
	

	Saturday_23_N

	TBA2
	MBB3293: errors on MM1 

· no temperature value

· no cmp signals

· bucking ratio too high

After the LHC cycle red alarms on the angle differences in the plateaus.
	MTM
	MMP
	Federico called stephane. MMP settings not done in cmp mode.

Temperature problem unsolved. Stephane advised to proceed with the tests

Stephane asked to stop the MM. Finish the test and remove the shafts.

	Saturday_23_N

	
	On SSS079  the shock-log box was not removed
	
	
	TC asked to remove the box.

	Sunday_24_M

	TBC2
	HV test at cold  failed in MBBL1105 QH vs ground, capacitance value out of range for one QH
	MTM
	
	Federico asked to start the special procedure checking the QH one by one.

He phoned to A.Siemko for advice. He advised to go on with the test.

	Sunday_24_M

	TBA2
	Magnet has temperature rise after discharge of QH
	MTM
	
	Federico contacted Giorgio. OK to continue the test. Check Monday.

	Sunday_24_N

	TBA1

TBE1
	PT8 generated in folder "MBBR_2_2_2_2”.

The file for Prep 2.1 (IAP@warm LF) was generated in 3317_2 folder even though it's not repeat magnet.
	AB/CO
	TEMA
	Routine corrected by Marco Monday Morning.

	Monday_25_M

	all
	No MTF
	TS/CE
	MTF
	Work with HRWREC off.


 










